Author’s response to reviews

Title: Concordance for Changes in Allergic Asthma Domain Variables after Short-term Corticosteroid Therapy

Authors:

Philip Silkoff (philsilkoff@gmail.com)
Mark Sarno (mjsarno@att.net)
Solomon Ssenyange (solomon@spirosure.com)
Vivek Balasubramanyam (vivek@spirosure.com)
Brian Awabdy (brian@spirosure.com)
Ryan Leard (Ryan@Spirosure.com)

Version: 2 Date: 15 Mar 2020

Author’s response to reviews:

Comment: minor revision concerning the required statistical analysis (FeNO changes before and after the treatment in atopic and non-atopic asthmatic patients) is mandatory.

Response: We have performed this analysis and added this under the results section for the primary endpoint. The allergic ITT population (n=76) showed an almost identical % change in FeNO after corticosteroids.